Author’s response to reviews

Title: Differential proteomic analysis of virus-enriched fractions obtained from plasma pools of patients with dengue fever or severe dengue

Authors:

Romain Fragnoud (romain.fragnoud@hotmail.fr)
Marie Flamand (Marie.flamand@pasteur.fr)
Frederic Reynier (Frederic.REYNIER@bioaster.org)
Philippe Buchy (pbuchy@pasteur-kh.org)
Duong Vasna (dvasna@pasteur-kh.org)
Alexandre Pachot (qllln2@yahoo.fr)
Glaucia Paranhos-Baccala (glaucia.baccala@fondation-merieux.org)
frederic Bedin (frederic.bedin@biomerieux.com)

Version: 2 Date: 19 Oct 2015

Author’s response to reviews:

Answers to the Editorial requests ( manuscript # INFD-D-15-00240)

1. Include all author e-mail addresses on the title page.

e-mail addresses have been included in the title page for all authors

2. Clarify in the Methods who granted the authors permission to use the samples.

A sentence has been added in the Methods that point out the people who granted to use the samples (see lines 127-130)

3. Abbreviations: Please include a list of abbreviations used in the manuscript and their meanings after the Conclusions section.

A list of abbreviation has been added a the end of the manuscript (see Abbreviations p.30)

4. Please include details of any funding in the Acknowledgement section.
The work was funded by BioMerieux SA. A sentence has been added in the Acknowledgement section (see line 683)

5. Upload a clean version of the manuscript without tracked changes.

A clean version of the manuscript has been uploaded.